Author's response to reviews

Title: Predictors of self-rated health. A 12-month prospective study of IT and media workers.

Authors:

Dan Hasson (dan.hasson@pubcare.uu.se)
Bengt B Arnetz (bengt.arnetz@pubcare.uu.se)
Tores Theorell (tores.theorell@ipm.ki.se)
Ulla Maria Anderberg (ullamaria.anderberg@pubcare.uu.se)

Version: 6 Date: 19 July 2006

Dear Population Health Metrics Editorial Team,

We are very happy for your confirmation that Population Health Metrics will publish our manuscript. In the final version the requested formatting changes have been done.

1. Qualifications have been removed from the Title page.
2. The Introduction section has been renamed 'Background'.
3. The running header has been removed from the Title page.
4. The shading has been removed from the Tables and lines thickened instead.
5. The manuscript has been checked for typing errors and a couple have been corrected.

With best wishes

Dan Hasson